Author's response to reviews

Title: Reproducibility and accuracy of optic nerve sheath diameter assessment using ultrasound compared to magnetic resonance imaging

Authors:

Jochen Baueuerle (jochen.baeuerle@uniklinik-freiburg.de)
Florian Schuchardt (florian.schuchardt@uniklinik-freiburg.de)
Laure Schroeder (laure.schroeder@uniklinik-freiburg.de)
Karl Egger (karl.egger@uniklinik-freiburg.de)
Matthias Weigel (matthias.weigel@uniklinik-freiburg.de)
Andreas Harloff (andreas.harloff@uniklinik-freiburg.de)

Version: 2 Date: 4 April 2013

Author's response to reviews:

Dear Ms. Cruz,

we have changed the manuscript accordingly to your advice.

We obtained written informed consent from all participants before they entered the study.

Our study was approved by the local ethics committee of the Albert Ludwigs University Freiburg, Germany.

Both statements are included in the method section and highlighted with yellow.

Best regards
Yours sincerely
Jochen Bäuerle